

Swiss Risk Management Plan Summary

TENKASI[®] (Oritavancin)

Document Version 1.0 (30.05.2022) Based on EU RMP (previously named Orbactiv) version 3.1 (signed on 22.09.2020)

Marketing authorization holder: A. Menarini GmbH, Switzerland Tenkasi[®] powder for concentrate for solution for infusion

The Risk Management Plan (RMP) is a comprehensive document submitted as part of the application dossier for market approval of a medicine. The RMP summary contains information on the medicine's safety profile and explains the measures that are taken in order to further investigate and follow the risks as well as to prevent or minimise them. The RMP summary of Tenkasi (previously Orbactiv) is a concise document and does not claim to be exhaustive. As the RMP is an international document, the summary might differ from the "Arzneimittelinformation / Information sur le médicament" approved and published in Switzerland, e.g. by mentioning risks occurring in populations or indications not included in the Swiss authorization. Please note that the reference document which is valid and relevant for the effective and safe use of Tenkasi in Switzerland is the "Arzneimittelinformation / Information sur le médicament" (see www.swissmedic.ch) approved and authorized by Swissmedic. A. Menarini GmbH is fully responsible for the accuracy and correctness of the content of the published summary RMP of Tenkasi (previously Orbactiv).

SUMMARY OF RISK MANAGEMENT PLAN FOR ORBACTIV (ORITAVANCIN)

This is a summary of the risk management plan (RMP) for Orbactiv. The RMP details important risks of Orbactiv, how these risks can be minimised, and how more information will be obtained about Orbactiv's risks and uncertainties (missing information).

Orbactiv's summary of product characteristics (SmPC) and its package leaflet give essential information to healthcare professionals and patients on how Orbactiv should be used.

This summary of the RMP for Orbactiv should be read in the context of all this information including the assessment report of the evaluation and its plain-language summary, all which is part of the European Public Assessment Report (EPAR).

Important new concerns or changes to the current ones will be included in updates of Orbactiv's RMP.

I. The medicine and what it is used for

Orbactiv is authorised for the treatment of acute bacterial skin and skin structure infections (ABSSSI) in adults (see SmPC for the full indication). It contains oritavancin as the active substance and it is given by intravenous infusion (400 mg of powder for concentrate for solution for infusion) administration.

Further information about the evaluation of Orbactiv's benefits can be found in Orbactiv's EPAR, including in its plain-language summary, available on the European Medicines Agency website, under the medicine's webpage link to the EPAR summary landing page.

II. Risks associated with the medicine and activities to minimise or further characterise the risks

Important risks of Orbactiv, together with measures to minimise such risks and the proposed studies for learning more about Orbactiv's risks, are outlined below.

Measures to minimise the risks identified for medicinal products can be:

- Specific information, such as warnings, precautions, and advice on correct use, in the package leaflet and SmPC addressed to patients and healthcare professionals;
- Important advice on the medicine's packaging;
- The authorised pack size the amount of medicine in a pack is chosen so to ensure that the medicine is used correctly;
- The medicine's legal status the way a medicine is supplied to the patient (e.g. with or without prescription) can help to minimise its risks.

Together, these measures constitute routine risk minimisation measures.

In addition to these measures, information about adverse reactions is collected continuously and regularly analysed, including PSUR assessment, so that immediate action can be taken as necessary. These measures constitute *routine pharmacovigilance activities*.

II.A. List of important risks and missing information

Important risks of Orbactiv are risks that need special risk management activities to further investigate or minimise the risk, so that the medicinal product can be safely administered. Important risks can be regarded as identified or potential. Identified risks are concerns for which there is sufficient proof of a link with the use of Orbactiv. Potential risks are concerns for which an association with the use of this medicine is possible based on available data, but this association has not been established yet and needs further evaluation. Missing information refers to information on the safety of the medicinal product that is currently missing and needs to be collected (e.g., on the long-term use of the medicine);

List of important risks and missing information	
Important Identified Risks	Hypersensitivity
Important Potential Risks	Pseudomembranous colitis / Clostridium difficile-associated diarrhea
	(CDAD)
	Osteomyelitis
Missing Information	None

II.B. Summary of important risks

Important Identified Risk: Allergic react	ions (Hypersensitivity)
Evidence for linking the risk to the medicine	Serious allergic reactions (rash, itching, redness, problems with breathing) have been seen following treatment with oritavancin. In a clinical study 7 out of 100 patients treated with oritavancin had an allergic reaction compared with 14 out of 100 patients who took vancomycin. Patients who had an allergic reaction following a glycopeptide other than oritavancin are likely to also be allergic to oritavancin (this is referred to as cross-sensitivity).
Risk factors and risk groups	Known hypersensitivity to this class of antibiotics.
Risk minimisation measures	Routine risk minimisation measures:
	 SmPC section 4.3 Contraindications SmPC section 4.4 Special warnings and precautions for use SmPC section 4.8 Undesirable effects
	The PL of the concerned products is in line with the information contained in the SmPC previously described. Such information is given in the following sections of the PL:
	 PL Section 2 What you need to know before you take You must not be given Warnings and precautions PL Section 4 Possible side effects
	Legal status: prescription only medicine
	Additional risk minimisation measures: No risk minimisation measures

Important Potential Risks: Antibiotic-associated diarrhea/infectious diarrhoea [Pseudomembranous colitis / Clostridium difficile-associated diarrhea (CDAD)]		
Evidence for linking the risk to the medicine	Antibiotic-associated diarrhoea refers to diarrhoea that develops in a person who is taking or recently took antibiotics. Some antibiotics can decrease the levels of protective bacteria normally found in the gut, and when this happens, harmful bacteria may be able to multiply and cause symptoms such as cramping pain, fever, and diarrhoea, sometimes occurring more than 2 months after receiving antibiotic	

	-
	treatment. One of the most serious causes of antibiotic associated diarrhoea is infection with a bacterium called Clostridium difficile. Antibiotic-associated diarrhoea has been reported with oritavancin. Patients who experience prolonged or severe diarrhoea following their treatment with oritavancin should contact their healthcare provider. Appropriate treatment should be considered.
Risk factors and risk groups	Patients with the following factors are at risk of antibiotic-associated diarrhea / infectious diarrhoea: 1) compromised immune system; 2) old age; 3) serious illness; 4) extended hospital stay; 5) extended course and / or multiple antibiotic treatment; 6) presence of a nasogastric tube; and 7) anti-ulcer medications.
Risk minimisation measures	 Routine risk minimisation measures: SmPC section 4.4 Special warnings and precautions for use SmPC section 4.8 Undesirable effects The PL of the concerned products is in line with the information contained in the SmPC previously described. Such information is given in the following sections of the PL: PL Section 2 What you need to know before you take Warnings and precautions PL Section 4 Possible side effects Legal status: prescription only medicine Additional risk minimisation measures: No risk minimisation measures

Important Potential Risks: Infection or inflammation of the bone or bone marrow (Osteomyelitis)		
Evidence for linking the risk to the medicine	More cases of osteomyelitis (infection in the bone) were reported with oritavancin than with vancomycin. Patients should be monitored for signs and symptoms of osteomyelitis following treatment with oritavancin. If osteomyelitis is diagnosed or suspected, alternative antibiotic treatment should be started. Oritavancin is not approved for the treatment of bone or bone marrow infections. Patients suspected or confirmed to have underlying bone or bone marrow infections should receive appropriate treatment.	
Risk factors and risk groups	During the clinical development, patients with diabetes mellitus at baseline had a higher rate of osteomyelitis than non-diabetic patients; the incidence of osteomyelitis was higher in oritavancin- treated patients with baseline peripheral vascular disease vs patients without peripheral vascular disease.	
Risk minimisation measures	 Routine risk minimisation measures: SmPC section 4.4 Special warnings and precautions for use SmPC section 4.8 Undesirable effects The PL of the concerned products is in line with the information contained in the SmPC previously described. Such information is given in the following sections of the PL: PL Section 2 What you need to know before you take Warnings and precautions PL Section 4 Possible side effects Legal status: prescription only medicine Additional risk minimisation measures: No risk minimisation measures 	

II.C. Post-authorisation development plan

II.C.1. Studies which are conditions of the marketing authorisation

There are no studies which are conditions of the marketing authorisation or specific obligation of Orbactiv.

II.C.2. Other studies in post-authorisation development plan

There are no studies required for Orbactiv.